Author’s response to reviews

Title: The Estimated Incidence of Induced Abortion in Kenya: a cross-sectional study

Authors:

Shukri F Mohamed (smohamed@aphrc.org)
Chima Izugbara (cizugbara@aphrc.org)
Ann M Moore (amoore@guttmacher.org)
Michael Mutua (mmutua@aphrc.org)
Elizabeth W Kimani-Murage (ekimani@aphrc.org)
Abdhalah K Ziraba (aziraba@aphrc.org)
Akinrinola Bankole (abankole@guttmacher.org)
Susheela D Singh (ssingh@guttmacher.org)
Caroline Egesa (cegesa@aphrc.org)

Version: 4 Date: 24 July 2015

Author’s response to reviews: see over
The Editor,
BMC pregnancy and Childbirth
July 23rd 2015

Dear Sir/Madam,


Thank you for the editorial comments. The authors have considered all the editors comments. The responses are below the comments and they are italicized for ease of following.

Editorial Comments:

1 - Please clarify the ethics approval statement in the manuscript

   • I hope the ethics statement in the manuscript is now clarified – see line 176-179.

2 - Please include a sentence as to whether consent to participate was obtained from the participants and if this was verbal or written.

   • A statement has been included to clarify the consent process. See line 228.

3 - Please replace the footnotes with endnotes

   • Noted and replaced

4 - Please rename Contribution to authorships ' Authors' contributions

   • Noted and changed has been done

We thank the editor for the above comments and clarifications on the paper.

Shukri F. Mohamed
Corresponding Author